6

https://doi.org/10.1038/s43856-024-00558-z

Differences in metalloproteinases and their tissue inhibitors in the cerebrospinal fluid are associated with delirium

Check for updates

Mari Aksnes [®] ¹ [⊠], Mari Haavig Schibstad², Farrukh Abbas Chaudhry³, Bjørn Erik Neerland⁴, Gideon Caplan^{5,6}, Ingvild Saltvedt^{7,8}, Rannveig S. Eldholm ^{® 7,8}, Marius Myrstad⁹, Trine Holt Edwin ^{® 10}, Karin Persson^{10,11}, Ane-Victoria Idland^{4,12}, Christian Thomas Pollmann¹³, Roy Bjørkholt Olsen¹⁴, Torgeir Bruun Wyller^{1,4}, Henrik Zetterberg ^{® 15,16,17,18,19,20}, Emma Cunningham ^{® 21} & Leiv Otto Watne ^{® 1,4,6,22}

Abstract

Background The aetiology of delirium is not known, but pre-existing cognitive impairment is a predisposing factor. Here we explore the associations between delirium and cerebrospinal fluid (CSF) levels of matrix metalloproteinases (MMPs) and their tissue inhibitors (TIMPs), proteins with important roles in both acute injury and chronic neurodegeneration.

Methods Using a 13-plex Discovery Assay[®], we quantified CSF levels of 9 MMPs and 4 TIMPs in 280 hip fracture patients (140 with delirium), 107 cognitively unimpaired individuals, and 111 patients with Alzheimer's disease dementia. The two delirium-free control groups without acute trauma were included to unravel the effects of acute trauma (hip fracture), dementia, and delirium.

Results Here we show that delirium is associated with higher levels of MMP-2, MMP-3, MMP-10, TIMP-1, and TIMP-2; a trend suggests lower levels of TIMP-4 are also associated with delirium. Most delirium patients had pre-existing dementia and low TIMP-4 is the only marker associated with delirium in adjusted analyses. MMP-2, MMP-12, and TIMP-1 levels are clearly higher in the hip fracture patients than in both control groups and several other MMP/TIMPs are impacted by acute trauma or dementia status.

Conclusions Several CSF MMP/TIMPs are significantly associated with delirium in hip fracture patients, but alterations in most of these MMP/TIMPs could likely be explained by acute trauma and/or pre-fracture dementia. Low levels of TIMP-4 appear to be directly associated with delirium, and the role of this marker in delirium pathophysiology should be further explored.

Delirium is characterised by acute, temporary disturbances in attention and cognition¹. This severe neuropsychiatric syndrome is a common complication of acute illness in hospitalised patients and a strong predictor of future cognitive decline and mortality^{2,3}. Older age and cognitive impairment are key predisposing factors for delirium^{1,4}. The aetiology of delirium is poorly understood, but it is proposed that delirium occurs when alterations in distinct neurobiological mechanisms increase the brain's vulnerability to acute triggers (e.g., surgery)^{1,5}.

Matrix metalloproteinases (MMPs) constitute a family of proteinases with central roles in extracellular matrix remodelling, acute and chronic neuroinflammation, and blood-brain barrier (BBB) permeability^{6,7}. MMPs Plain language summary

Delirium is a syndrome in which there are substantial changes in a person's ability to focus, understand, or pay attention to events. Delirium often occurs in response to sudden trauma and is more common in persons with pre-existing cognitive impairment. What happens in the brain during delirium is not well understood. To learn more, we have studied whether markers in the cerebrospinal fluid were altered in people with delirium compared to people without delirium. To understand differences specifically caused by delirium, we included two control groups without acute trauma, one with cognitively healthy participants and one with dementia patients. We found several markers altered in people with delirium, with most of the markers similarly altered in people with cognitive impairment due to dementia. One marker was directly linked to delirium and could potentially shed light on the brain processes that cause the syndrome.

cleave several different substrates including extracellular matrix components, signalling molecules, and inflammatory mediators such as cytokines. Due to their many proteolytic substrates, high MMP activity can be detrimental and is therefore tightly regulated by tissue inhibitors of metalloproteinases (TIMPs)⁸. Expression of different TIMPs has a degree of organ specificity, and TIMP-4 exhibits greater specificity for the brain⁹. Altered MMP activity has been linked to both progressive neurodegenerative diseases such as Alzheimer's disease (AD) and acute neurological injury such as ischemic stroke^{10,11}. In aged mice and rats, peripheral surgery appears to induce BBB dysfunction and elevate hippocampal expression of MMP-2 and MMP-9¹²⁻¹⁴. During experimentally induced white matter pathology

A full list of affiliations appears at the end of the paper. De-mail: mari.aksnes@medisin.uio.no

levels of TIMP-4 are supressed; this may be important for delirium given evidence of altered white matter connectivity on MRI studies and elevated levels of neurofilament light in blood and cerebrospinal fluid (CSF) after delirium¹⁵⁻¹⁷. Lower plasma levels of MMP-9 have been associated with delirium in critically ill patients¹⁸. Associations with other MMPs and their inhibitors have not been explored and the understanding of how different MMPs and TIMPs might interact in delirium is limited. Furthermore, the association between CSF MMP and TIMP levels and delirium has not been explored.

The primary aim of this study was to determine whether delirium is associated with alterations in CSF MMP and TIMP levels. We investigated thirteen MMPs and TIMPs in acutely hospitalised patients with hip fracture, with or without delirium and/or dementia. Further, secondary aims were to investigate whether CSF MMPs and TIMPs were connected to any specific clinical aspect of delirium (e.g., incident or prevalent delirium, biomarkers of neuronal injury, mortality) and to disentangle the effects of acute trauma, delirium, and dementia by comparison with two delirium-free control groups without acute trauma (a group of cognitively unimpaired (CU) controls and a group of persons with AD dementia). We show that while several MMPs and TIMPs are altered in delirium, only TIMP-4 is decreased independent of pre-existing cognitive impairment.

Methods

Cohorts

This was a multicentre observational study including patients from four hospitals in the Oslo Region, Norway and one hospital in Trondheim, Norway. Patients were included between 2009 and 2019, and CSF samples were available from 280 hip fracture patients (140 with delirium), 107 CU controls, and 111 patients with AD dementia.

Hip fracture cohort. The hip fracture patients were included in a multicentre study run at Oslo University Hospital, Diakonhjemmet Hospital, Akershus University Hospital, and Bærum Hospital between 2016 and 2019. All patients admitted for surgical repair of their hip fracture in spinal anaesthesia were eligible for inclusion; informed consent was obtained from all patients or, in the presence of cognitive impairment, from the next of kin. Delirium was assessed according to the DSM-5 criteria based on a standardised procedure. In short, delirium was assessed bedside daily, with tests of cognition and attention¹⁹⁻²¹ in all patients until the fifth postoperative day or until discharge in patients with delirium. Patients with delirium were further divided into prevalent delirium (those with delirium at the time of CSF sampling) and incident delirium (those without delirium at the time of CSF sampling who later developed it). In patients not fulfilling all criteria for delirium, subsyndromal delirium was defined as evidence of cognitive change together with any of the following: altered arousal; attentional deficits; other cognitive change; or delusions or hallucinations²². Study nurses, trained in delirium assessment by the study physician (LOW), performed all assessments. Two experienced delirium researchers (LOW and BEN) independently assessed all available information for each patient to decide whether the DSM-5 criteria for delirium were fulfilled or not. The interrater agreement upon delirium diagnosis was excellent (kappa 0.97), with disagreements resolved through discussion. Pre-fracture cognitive status was assessed with the Informant Questionnaire on Cognitive Decline in the Elderly (IQCODE); scores \geq 3.44 were considered to indicate dementia²³. In the case of missing IQCODE scores (n = 21), prefracture dementia status was established retrospectively using hospital records. The preoperative American Society of Anesthesiologists (ASA) physical status classification was used as a measure of medical comorbidities²⁴.

Cognitively unimpaired control group (CU group). Cognitively unimpaired individuals aged 65 years or older were recruited from the Cognorm study of patients undergoing elective gynaecological, orthopaedic or urological surgery in spinal anaesthesia at Oslo University Hospital or Diakonhjemmet Hospital between 2012 and 2013²⁵. All patients provided informed consent. The CU group was tested at baseline and at yearly follow-up visits with a standardised battery of cognitive tests²⁵, and only individuals considered cognitively unimpaired at baseline in line with criteria employed in Knapskog et al.²⁶ were included. All CU controls were free from delirium at the time of CSF sampling. Patient journals were examined to evaluate incident delirium in the CU group; two patients were determined to have post-operative delirium.

Alzheimer's disease dementia control group (AD group). Dementia is a primary risk factor for delirium, and as such there is high overlap between patients with dementia and patients who develop delirium⁴. To disentangle these effects in our study, we included an AD dementia control group (AD group). These patients were assessed at two Norwegian outpatient memory clinics, Oslo University Hospital, and St. Olav University Hospital, and included in the Norwegian Registry of Persons Assessed for Cognitive Symptoms (NorCog) between 2009 and 2018; all patients provided informed consent. Patients were assessed according to a standardised research protocol by experienced memory clinic physicians²⁷. All patients met the criteria for probable or possible dementia due to AD²⁸ and had pathological levels of the AD CSF biomarkers, CSF amyloid- β_{42} (A β_{42}) and phosphorylated tau₁₈₁ (p-tau₁₈₁)²⁹. CSF sampling was conducted as part of the diagnostic procedure and was not performed in the presence of precipitating factors for delirium such as sepsis, hip fracture, or other acute illness; all patients in the AD group were free from delirium at the time of CSF sampling and no anaesthetic agent was administered before lumbar puncture.

CSF sampling and biochemical analysis

In the hip fracture patients and CU group, CSF was collected at the onset of spinal anaesthesia, before anaesthetic agents were administered. In the AD group, CSF was collected as part of the clinical work up for assessment of potential dementia. In all patients, CSF was collected in sterile polypropylene tubes and centrifuged for 10 minutes at 2000 G. The samples were aliquoted into 0.5 ml polypropylene tubes and stored at -80 °C. All CSF samples were sent on dry ice for biochemical analyses to the Eve Technologies laboratory (Calgary, Canada) where MMPs and TIMPs were analysed using a 13-plex Discovery Assay® on a Luminex® xMAP® instrument. The assay simultaneously measures MMP-1, MMP-2, MMP-3, MMP-7, MMP-8, MMP-9, MMP-10, MMP-12, MMP-13, TIMP-1, TIMP2-, TIMP-3 and TIMP-4 in a single microwell. Samples were measured in duplicate. Eve Technologies regularly performs quality controls to ensure that inter-assay variability falls within the range of 5–20%. CSF $A\beta_{42}$ and p-tau₁₈₁ concentrations for the hip fracture cohort were measured using INNOTEST enzyme-linked immunosorbent assays (Fujirebio) at Sahlgrenska University Hospital (Mölndal, Sweden).

Statistics and reproducibility

Statistical analyses were performed in STATA 16.1 and data visualisations were created in R4.1.1 using RStudio. Categorical variables were compared using the χ^2 test. Continuous variables were compared using the Mann-Whitney U non-parametric test, as several variables were not normally distributed. We investigated correlations between variables using Spearman's rho correlation. All reported *P*-values are two-sided and due to multiple comparisons only *P*-values less than <0.01 were considered statistically significant.

To investigate the effects of MMPs/TIMPs on delirium we performed univariable and multivariable logistic regression analyses in the hip fracture cohort with delirium as the dependent variable and each measured MMP/ TIMP as the independent variable. Moreover, within the delirium patients, we performed multivariate logistic regression analyses for each measured MMP/TIMP with specific features of delirium (presence/absence of hallucinations, illusions, and motoric restlessness) as the dependent variable. To explore the association between the MMPs/TIMPs and 1-year mortality following hip fracture, we performed survival analysis using Cox regression in the hip fracture patients, censored 365 days after admission to the hospital. All multivariate analyses (logistic and Cox regression) were controlled for sex, age, and dementia status (IQCODE \geq 3.44 = dementia, IQCODE < 3.44 = no dementia) as these factors are known to influence CSF biomarkers, delirium risk, or both. The survival analysis was adjusted for delirium (presence/absence). One hip fracture patient had extreme values (>15 standard deviations above the mean) of several MMPs and was excluded from further analysis.

MMP-1, MMP-7, MMP-8, MMP-9, and MMP-13 were detectable in less than 50% of samples and were therefore excluded from further analysis. In a minority of samples, MMP-10 (2.4%) and MMP-12 (26.4%) were below the lower limit of quantification (LLQ); for MMP-10 and MMP-12 values below the LLQ were replaced with values randomly drawn from a uniform distribution with a minimum value of 0 and a maximum value of the respective LLQs. MMP-2, MMP-3, and all TIMPs were detectable in all samples.

Reporting summary

Further information on research design is available in the Nature Portfolio Reporting Summary linked to this article.

Results

MMP and TIMP levels in the hip fracture patients

The effect of delirium status on MMP/TIMP levels. The characteristics of the hip fracture patients stratified by delirium status are presented in Table 1. The patients with delirium were significantly older (median age 87 years) than the patients without delirium (median age 77 years, $P = 3.7*10^{-12}$). The delirium patients also had significantly higher scores on the IQCODE ($P = 2.0*10^{-15}$) and more severe medical comorbidities with 65.5% ASA-score III-IV versus 38.8% in the patients without delirium, $P = 1.4*10^{-14}$.

Several MMPs and TIMPs differed significantly between the no delirium and delirium groups, see Table 1. MMP-2, MMP-3, MMP-10, TIMP-1, and TIMP-2 were higher in the delirium group, and there was a trend (P = 0.01) suggesting lower TIMP-4 in patients with delirium compared to patients without delirium. The distribution of CSF MMP and TIMP levels in the delirium and non-delirium group are presented in Fig. 1.

Within the delirium group, there were no differences in MMP or TIMP levels between patients with the incident (n = 73) and prevalent delirium (n = 67, all P > 0.05). Patients with subsyndromal delirium (n = 21) were

included in the no-delirium group; the results were not impacted by the exclusion of these patients in sensitivity analyses.

The effect of dementia status on MMP and TIMP levels. The characteristics of the hip fracture patients stratified by dementia status are presented in Supplementary Table 1. The dementia group was significantly older (median age 86 years) than the no dementia group (median age 79.5 years, $P = 9.4*10^{-6}$). Delirium occurred more commonly in the dementia patients (83.9%) than in the non-dementia patients (29.3%, $P = 5.2*10^{-22}$). Compared to hip fracture patients without dementia, the dementia group had higher levels of MMP-10 (median 33.7 pg/mL vs. 20.4 pg/mL, $P = 9.4*10^{-6}$).

The interplay of dementia and delirium. As there is a large overlap between hip fracture patients with dementia and with delirium, we further investigated MMP and TIMP levels by stratifying patients by dementia and delirium status. Patient characteristics stratified by dementia and delirium status are presented in Supplementary Table 2.

In patients without dementia, those with delirium (n = 41) had significantly higher levels of MMP-3, MMP-10, and TIMP-2 compared to those without (n = 120), see Supplementary Table 3. In the patients with dementia, there were no significant differences in MMP and TIMP levels associated with delirium status.

There were no significant differences between patients with incident delirium compared to patients with prevalent delirium, regardless of dementia status, see Supplementary Table 3.

MMP and **TIMP** levels as predictors of delirium. In univariate logistic regressions, higher levels of MMP-2, TIMP-1, and TIMP-2 significantly increased the odds of delirium, see Table 2. In analyses adjusted for sex, age, and cognitive impairment, low TIMP-4 was the only MMP/TIMP significantly associated with delirium. In patients with delirium, none of these markers were associated with increased odds of specific features such as hallucinations (n = 27 patients, 22.7%, 21 missing), illusions (n = 24 patients, 19.8%, 19 missing) or motoric restlessness (n = 66 patients, 51.6%, 12 missing) in multivariate logistic regression analyses, see Supplementary Table 4.

MMP and TIMP associations with 1-year mortality following hip fracture. Among the 279 hip fracture patients, there were 59 (21%)

Table 1 | Characteristics of the hip fracture cohort stratified by delirium-status

	All	No delirium	Delirium	P
Ν	279	139	140	
Age	84 (74; 89)	77 (69; 86)	87 (81; 91)	3.7*10 ⁻¹²
Female sex, n (%)	192 (68.8)	99 (71.2)	93 (66.4)	0.37
Dementia, n (%)	118 (42.1)	19 (16.1)	99 (83.9)	5.2*10 ⁻²²
IQCODE	3.3 (3.0; 4.2)	3.1 (3.0; 3.3)	3.9 (3.4; 4.8)	2.0*10 ⁻¹⁵
ASA III-IV, n (%)	145 (52.0)	54 (38.8)	91 (65.0)	1.4*10 ⁻¹⁴
MMP-2 (ng/mL)	59.3 (49.3; 69.4)	56.8 (47.3; 65.0)	62.2 (52.5; 73.5)	0.001
MMP-3 (pg/mL)	227 (154; 335)	205 (147; 299)	251 (177; 360)	0.008
MMP-10 (pg/mL)	24.2 (13.5; 41.3)	19.2 (10.3; 31.8)	32.4 (17.9; 45.7)	4.7*10 ⁻⁶
MMP-12 (pg/mL)	3.6 (0.4; 6.7)	3.6 (0.4; 7.8)	3.6 (0.4; 6.7)	0.59
TIMP-1 (ng/mL)	87.3 (66.3; 110.0)	82.3 (60.5; 103.3)	92.1 (72.7; 121.0)	0.001
TIMP-2 (ng/mL)	80.4 (69.0; 93.7)	76.2 (66.1; 87.3)	85.7 (71.1; 99.3)	1.2*10 ⁻⁴
TIMP-3 (ng/mL)	15.7 (14.4; 17.0)	15.6 (14.1; 16.7)	15.9 (14.4; 17.4)	0.03
TIMP-4 (ng/mL)	1.6 (1.3; 1.9)	1.6 (1.4; 2.0)	1.5 (1.2; 1.9)	0.01

Data is presented as median (quartile 1; quartile 3) unless otherwise indicated. P value is for comparison between patients with and without delirium using Mann Whitney U test (continuous variables) or χ^2 -test (categorical variables).

ASA American Society of Anesthesiologists physical status classification, IQCODE Informant Questionnaire on Cognitive Decline in the Elderly, MMP matrix metalloproteinase, TIMP tissue inhibitor of matrix metalloproteinase.



Fig. 1 | Distributions of cerebrospinal fluid MMP and TIMP levels across the hip fracture patients with no delirium (n = 139) and with delirium (n = 140). The violin plots show the entire distribution of each marker, whilst the inserted box plot shows the median (middle line), 1st quartile (lower box limit) and 3rd quartile (upper box limit). Black lines show the minimum (minimum data value – 1.5*interquartile

range) and maximum (maximum data value + 1.5* interquartile range). Black circles show outliers. *MMP* matrix metalloproteinase, *TIMP* tissue inhibitor of matrix metalloproteinase.

Table 2 | Univariate and adjusted logistic regression models predicting odds of delirium in the hip fracture cohort (n = 279)

	Univariate		Adjusted*	
	Odds ratio (95% CI)	Р	Odds ratio (95% CI)	Р
MMP-2	1.025 (1.009; 1.041)	0.002	1.006 (0.988; 1.025)	0.51
MMP-3	1.001 (0.999; 1.002)	0.11	1.000 (0.999; 1.001)	0.98
MMP-10	1.011 (1.002; 1.021)	0.02	1.002 (0.993; 1.011)	0.65
MMP-12	0.977 (0.924; 1.032)	0.40	0.948 (0.883; 1.016)	0.13
TIMP-1	1.013 (1.005; 1.021)	0.001	1.003 (0.992; 1.014)	0.59
TIMP-2	1.026 (1.013; 1.039)	8.5*10 ⁻⁵	1.015 (0.998; 1.031)	0.08
TIMP-3	1.164 (1.029; 1.318)	0.02	1.020 (0.861; 1.205)	0.82
TIMP-4	0.550 (0.335; 0.902)	0.02	0.424 (0.224; 0.802)	0.008

*Adjusted for sex, age, and presence/absence of dementia (IQCODE ≥ 3.44 = dementia (n = 118), IQCODE < 3.44 = no dementia (n = 161). In all models, higher age and the presence of dementia were significant predictors of delirium.

CI confidence interval, IQCODE Informant Questionnaire on Cognitive Decline in the Elderly, MMP matrix metalloproteinase, TIMP tissue inhibitor of matrix metalloproteinase

deaths in the first year following surgery; 52 (37%) in the delirium group, and 7 in the no-delirium group (5%). In univariate analyses, several MMPs and TIMPs were associated with increased mortality; however, no MMPs or TIMPs were associated with mortality in analyses adjusted for age, sex, delirium, and pre-existing dementia, see Supplementary Table 5.

Cohort comparisons

We compared the levels of CSF MMP and TIMP in the hip fracture patients with two control groups, the CU group and the AD group; these control groups permitted us to explore how CSF MMP/TIMPs are affected by acute trauma (hip fracture group vs. groups with no acute illness), dementia, and Table 3 | Characteristics of the cognitively unimpaired group, the Alzheimer's disease group and the hip fracture patients

	1. CU group	2. AD group	3. Hip fracture	<i>P</i> 1. vs 3.	P 2. vs. 3
Ν	107	111	279 ^a		
Age	71 (67; 76)	71 (66; 75)	84 (74; 89)	1.4*10 ⁻¹⁸	2.5*10 ⁻²⁴
Female sex	48 (44.9)	68 (68.3)	192 (68.8)	1.4*10 ⁻⁵	0.15
Delirium	2 (1.9)	0 (0)	140 (50.0)		
Dementia ^b	0 (0)	111 (100.0)	118 (42.1)		
IQCODE	3 (3.0; 3.1)	3.8 (3.5; 4.1)	3.3 (3.0; 4.2)	4.9*10 ⁻¹⁹	4.8*10 ⁻⁶
ASA III-IV, n (%)	NA	NA	145 (52.0)		

Data are presented as *n* (%) for female sex, delirium, and dementia and as median (quartile 1; quartile 3) for age and IQCODE. *P*-values are for Mann Whitney U test (continuous variables) and χ^2 -test (categorical variables).

^aOne extreme outlier was excluded.

^bDementia in hip fracture cohort defined as IQCODE ≥ 3,44. In the case of missing IQCODE scores (*n* = 21), dementia status was established retrospectively using hospital records.

AD Alzheimer's disease, ASA American Society of Anesthesiologists physical status classification, CU cognitively unimpaired, IQCODE Informant Questionnaire on Cognitive Decline in the Elderly.

delirium. The cohort characteristics are presented in Table 3. The hip fracture patients were significantly older (median age 84 years) than both the CU group (median age 71 years, $P = 1.4*10^{-18}$) and the AD group (median age 71 years, $P = 2.5*10^{-24}$). There was a higher proportion of women in the hip fracture cohort (68.8%) compared to the CU group (44.9%). In the CU group, two patients were diagnosed with incident delirium following surgery based on journal inspection. Sensitivity analyses excluding these patients did not affect the results.

The distribution of CSF MMP and TIMP levels across the three different cohorts is presented in Fig. 2.

Most of the measured MMP/TIMPs differed significantly between the hip fracture patients and the control groups, suggesting that CSF MMP/ TIMP levels are affected by the acute trauma, see Table 4 for all comparisons. The markers that were most clearly elevated in the hip fracture patients compared to the other two groups (that did not experience acute trauma) were MMP-2 (~30% higher), MMP-12 (140% higher), and TIMP-1 (>50% higher).

MMP-3 and MMP-10 were the markers most clearly impacted by dementia. Compared to the CU group, the hip fracture cohort (in which 40% had dementia) had 15% higher levels of MMP-3 and 90% higher levels of MMP-10. For both markers, the levels in the AD group were even higher.

The effect of sex and age

In the hip fracture cohort, there was no difference in the frequency of delirium across men and women. Men had significantly higher levels of MMP-3, MMP-10, and TIMP-2, see Supplementary Table 6.

In all patients combined, age was moderately correlated with MMP-2 levels (rho = 0.58), TIMP-1 levels (rho = 0.54), and TIMP-2 levels (rho = 0.45). Age was weakly correlated with MMP-3 (rho = 0.22) and MMP-10 (rho = 0.28). In the three cohorts, age appeared most strongly correlated with MMP and TIMP levels in the hip fracture patients, and the pattern of correlation was mostly similar across the CU controls and the hip fracture patients. In the AD group, age was only weakly correlated with MMP-2, TIMP-1 and TIMP-2 levels. Correlations between age and all MMPs/TIMPs across the three cohorts are presented in Supplementary Table 7.

Associations with biomarkers of Alzheimer's disease pathophysiology

In the hip fracture cohort, several of the measured MMPs and TIMPs were weakly to moderately correlated with CSF p-tau₁₈₁ and NFL, see Supplementary Table 8. The strongest correlations were between MMP-10 and p-tau₁₈₁ (rho = 0.46) and MMP-2 and NFL (rho = 0.64). Notably only TIMP-4 was moderately correlated with CSF A β_{42} (rho = 0.41).

Discussion

We have quantified the levels of several MMPs and TIMPs in the CSF of patients with delirium. In a large cohort of hip fracture patients, we found that nearly all measured MMPs and TIMPs differed significantly between patients with and without delirium; in delirium patients, the levels of MMP-2, MMP-3, MMP-10, TIMP-1, and TIMP-2 were higher, and there was a trend suggesting TIMP-4 was lower. However, most of the associations between delirium and MMPs/TIMPs could likely be explained by the acute trauma of the hip fracture and/or pre-fracture dementia. In adjusted analyses, low TIMP-4 was the only marker that significantly increased the odds of delirium. However, neither TIMP-4 nor any of the measured MMP/TIMPs were associated with 1-year mortality following hip fracture.

To our knowledge, this is the first study to investigate CSF MMPs and TIMPs in association with delirium. MMPs/TIMPs have several roles in the central nervous system and the periphery, amongst them the regulation of neuroinflammation³⁰. Other markers of neuroinflammation such as CSF interleukin-8 have previously been linked to delirium³¹. However, when stratifying patients by dementia status, most of the group differences were attenuated, suggesting that they could be explained by the prevalence of dementia rather than delirium status. This highlights the importance of documenting pre-existing cognitive impairment in delirium biomarker studies and is a major strength of our study. We did not find any differences in CSF MMP/TIMPs between incident versus prevalent delirium.

We included two control groups, one group with cognitively unimpaired individuals and one with AD dementia patients, permitting us to explore the impact of the acute trauma of hip fracture, dementia, and delirium on MMP/TIMP levels. This is relevant as MMP/TIMPs are highly dynamic systems that are both activated during acute illness and impacted by dementia disorders¹⁰. Similarly, delirium is precipitated by acute illness and dementia is a key predisposing factor^{1,4}. It is therefore a key challenge in delirium biomarker studies to disentangle the effects of acute illness, dementia, and delirium.

Most of the measured MMP/TIMPs differed significantly between the hip fracture patients and the control groups, suggesting that CSF MMP/ TIMP levels are affected by acute trauma. The markers that were most clearly elevated in the hip fracture patients were MMP-2, MMP-12, and TIMP-1. It is well established that MMPs and TIMPs play important roles in response to acute traumas such as skin wounds, lung injury, and cardiovascular insults³²⁻³⁴. For example, MMP-2 expression is quickly increased in response to oxidative stress, and TIMP-1 plays essential roles in heart tissue remodelling³⁴. Similarly, MMPs are upregulated in the central nervous system in response to acute insults such as traumatic brain injury, brain haemorrhage, or ischaemic stroke⁸. Increased expression of MMP-12 is seen both after intracerebral haemorrhage and spinal cord injury in animal models^{35,36}; this upregulation is thought to have adverse effects. However, little is known about the effect of acute bodily trauma such as a hip fracture on the levels of these markers in the CSF or the central nervous system more broadly. Our results suggest that CSF MMP-2, MMP-12, and TIMP-1 are elevated in the CSF after hip fracture, but further research is needed to determine whether this is due to increased expression by neural and



Fig. 2 | Distributions of cerebrospinal fluid MMP and TIMP levels across the cognitively unimpaired group (n = 107), the Alzheimer's disease group (n = 111) and the hip fracture patients (n = 279). The violin plots show the entire distribution of each marker, whilst the inserted box plot shows the median (middle line), 1st quartile (lower box limit), and 3rd quartile (upper box limit). Black lines show the minimum

(minimum data value -1.5*interquartile range) and maximum (maximum data value +1.5* interquartile range). Black circles show outliers. AD Alzheimer's disease, CU cognitively unimpaired, MMP matrix metalloproteinase, TIMP tissue inhibitor of matrix metalloproteinase.

endothelial cells, or increased influx of neutrophils from the periphery⁸. Furthermore, as MMPs have both beneficial and detrimental roles in the central nervous system, it remains to be determined how, if at all, this upregulation contributes to the pathophysiology of delirium.

The levels of MMP-3 and MMP-10 were clearly higher in the hip fracture cohort, in which 40% of patients had dementia, compared to the CU controls, and even higher in the AD dementia cohort. This, along with the modest correlations with age, suggests that elevated levels of MMP-3 and MMP-10 are associated with dementia. This finding is in line with previous research linking increased MMP-10 levels to an increased risk of dementia and accelerated disease progression in neurodegenerative diseases^{37,38}. Similarly, CSF levels of MMP-3 have previously been linked to dementia and specifically AD pathology³⁹⁻⁴¹. In hip fracture patients, higher levels of MMP-3 and MMP-10 were associated with delirium, but only in those without pre-fracture dementia. This suggests that the expression of MMP-3 and MMP-10 in the CSF after acute injury and in the presence of delirium might be moderated by preexisting dementia disorders, highlighting the complex interplay between delirium and dementia⁴. Chronic neuroinflammation and changes in the balance between MMPs and TIMPs in dementia disorders might attenuate any response, such as e.g., an increase in MMP-3, to acute injury in dementia patients. In dementia-free patients, it is possible that higher levels of MMP-3 and MMP-10 indicate reduced brain resilience due to ageing or pre-symptomatic neurodegenerative disease, thus predisposing these patients to delirium.

In analyses adjusted for sex, age, and pre-existing cognitive impairment, only lower levels of TIMP-4 remained a significant predictor for delirium. TIMP-4 is the most recently discovered and least studied member of the TIMP-family⁴². TIMP-4 is generally thought to exert beneficial effects by contributing to extracellular matrix remodelling⁴³. CSF levels of TIMP-4 have primarily been studied in patients with bacterial or viral infections; lower levels of CSF TIMP-4 are found in patients with eosinophilic meningitis, syphilis, or HIV-1 and syphilis co-infections^{44,45}. One study also found reduced levels of TIMP-4 in the CSF of patients with hydrocephalus⁴⁶. The mechanisms driving the reduced expression of TIMP-4 in these conditions are not established, but it is speculated that alterations in the inflammatory milieu and blood-brain barrier damage are contributing factors⁴⁴⁻⁴⁶. In terms of acute effects, there is some indication that TIMP-4 expression should increase in response to trauma, as pericardial TIMP-4 concentrations increase significantly after cardiopulmonary bypass surgery⁴⁷. Interestingly, in our study, TIMP-4 levels did not differ significantly between the CU control group and the delirium patients.

Table 4 MMP an	d TIMP levels in th	e CU groups, AD gi	roup, and hip fracture I	oatients without and	ł with delirium				
	1. CU group	2. AD group	Hip fracture patients			Р			
			3. All	4. No delirium	5. Delirium	1. vs 3.	2. vs. 3	2. vs 4.	2. vs 5.
MMP-2 (ng/mL)	45.7 (39.5; 52.7)	44.1 (38.0; 50.2)	59.3 (49.3; 69.4)	56.8 (47.3; 65.0)	62.2 (52.5; 73.5)	$5.4^{*}10^{-21}$	1.2*10 ⁻²⁴	$2.8*10^{-24}$	$3.2^{*}10^{-15}$
MMP-3 (pg/mL)	197 (147; 266)	252 (182; 325)	227 (154; 335)	205 (147; 299)	251 (177; 360)	0.01	0.24	0.01	0.68
MMP-10 (pg/mL)	12.8 (6.5; 21.7)	29.3 (21.7; 38.1)	24.2 (13.5; 41.3)	19.2 (10.3; 31.8)	32.4 (17.9; 45.7)	2.7*10 ⁻⁹	0.06	2.1*10	0.37
MMP-12 (pg/mL)	1.5 (0.2; 5.7)	1.5 (0.3; 4.6)	3.6 (0.4; 6.7)	3.6 (0.4; 7.8)	3.6 (0.4; 6.7)	0.004	8.5*10 ⁻⁴	0.003	0.003
TIMP-1 (ng/mL)	57.4 (50.3; 70.1)	57.6 (49.5; 68.4)	87.3 (66.3; 110.0)	82.3 (60.5; 103.3)	92.1 (72.7; 121.0)	$3.8*10^{-18}$	2.6*10 ⁻¹⁹	1.4*10 ⁻¹¹	$1.2^{*}10^{-19}$
TIMP-2 (ng/mL)	67.1 (61.1; 73.7)	71.1 (65.3; 78.9)	80.4 (69.0; 93.7)	76.2 (66.1; 87.3)	85.7 (71.1; 99.3)	2.3*10 ⁻¹³	3.7*10 ⁻⁶	0.02	$3.1*10^{-9}$
TIMP-3 (ng/mL)	16.1 (15.4; 17.1)	16.4 (15.6; 17.3)	15.7 (14.4; 17.0)	15.6 (14.1; 16.7)	15.9 (14.4; 17.4)	0.003	5.3*10 ⁻⁵	9.0*10 ⁻⁷	0.03
TIMP-4 (ng/mL)	1.5 (1.2; 1.8)	1.7 (1.5; 2.0)	1.6 (1.3; 1.9)	1.6 (1.4; 2.0)	1.5 (1.2; 1.9)	0.27	0.007	0.29	1.8*10 ⁻⁴
Data is presented as media AD Alzheimer's disease. CU	n (quartile 1; quartile 3). <i>P</i> -valt / cognitively unimpaired. <i>MM</i>	ues are for Mann-Whitney U te ^o matrix metalloproteinase. <i>Til</i>	sst. WP tissue inhibitor of matrix metallor	oroteinase.					

Moreover, in the hip fracture patients TIMP-4 levels were positively associated with CSF A β_{42} ; low CSF A β_{42} is indicative of Alzheimer's disease pathology and known to predict postoperative delirium⁴⁸. We speculate that low TIMP-4 in delirium patients could indicate an insufficient response to acute damage and predispose these patients to delirium, especially in the cases of pre-existing amyloid pathology, but more research is needed to elucidate the contributions of TIMP-4 to delirium pathophysiology.

A major strength of the current study is the large CSF dataset on hip fracture patients with and without delirium. The inclusion of the two control groups, the CU group and the AD group, provides important context for the interpretation of our data. One limitation is the group differences in terms of CSF collection procedures and storage. Importantly, time in room temperature is known to influence CSF MMP levels⁴⁹, but CSF was rapidly frozen after collection in all cohorts. However, biobank storage time differed across the three cohorts and may have influenced the results as the effects of long-term storage on CSF MMP/TIMP levels are unknown. Delirium was assessed daily by trained investigators using validated instruments, and the diagnostic algorithm is documented in detail⁵⁰. Furthermore, we have included information on dementia status for all the hip fracture patients, which is essential as dementia is associated with both delirium⁴ and MMPs/TIMPs^{10,51,52}. However, due to the patients' acute admission, dementia status was determined using the IQCODE, which, whilst validated and commonly used⁵³, is not a substitute for objective cognitive testing or a complete dementia assessment. We have limited information on the underlying causes of dementia in the dementia group, and these patients might suffer from several interacting pathologies. Moreover, it is possible that the non-dementia group also contained people with undiagnosed underlying neurodegenerative diseases. As much of the previous research on MMPs/TIMPs investigated circulating levels of these markers, it would have strengthened our study to include paired CSF and blood samples.

In conclusion, CSF levels of MMP/TIMPs are altered in hip fracture patients with delirium compared to patients without delirium. Most of these markers can be linked to the acute insult (i.e., the hip fracture) and/or preexisting dementia in the delirium patients, but TIMP-4 was directly connected to delirium. This study illustrates the importance of collecting information on dementia and including relevant comparison groups in delirium biomarker studies. Further studies should explore the contribution of this marker to delirium pathophysiology.

Data availability

The numerical data underlying Figs. 1 and 2 can be found in Supplementary Data 1. The further data that support the findings of this study are not openly available in order to preserve the privacy of individual participants under the European General Data Protection Regulation (GDPR). Data are, however, available from the authors upon reasonable request contingent on approval from the ethical committee REC South East (contact at e-mail post@helseforskning.etikkom.no) and for the Alzheimer's disease patients the Norwegian Registry of Persons Assessed for Cognitive Symptoms (NorCog, contact at e-mail: post@aldringoghelse.no).

Received: 9 October 2023; Accepted: 20 June 2024; Published online: 27 June 2024

References

- 1. Wilson, J. E. et al. Delirium. Nat. Rev. Dis. Prim. 6, 90 (2020).
- Witlox, J. et al. Delirium in elderly patients and the risk of postdischarge mortality, institutionalization, and dementia: a metaanalysis. *JAMA* **304**, 443–451 (2010).
- Goldberg, T. E. et al. Association of delirium with long-term cognitive decline: a meta-analysis. *JAMA Neurol.* 77, 1373–1381, (2020).
- Fong, T. G. & Inouye, S. K. The inter-relationship between delirium and dementia: the importance of delirium prevention. *Nat. Rev. Neurol.* 18, 579–596 (2022).

- Yang, T., Velagapudi, R. & Terrando, N. Neuroinflammation after surgery: from mechanisms to therapeutic targets. *Nat. Immunol.* 21, 1319–1326 (2020).
- Okada, Y. Proteinases and Matrix Degradation. In *Kelley and Firestein's Textbook of Rheumatology*, 10th edn Vol. 1 (eds Gary S. Firestein et al.) 106–125 (Elsevier, 2017).
- Rivera, S., Khrestchatisky, M., Kaczmarek, L., Rosenberg, G. A. & Jaworski, D. M. Metzincin Proteases and their inhibitors: foes or friends in nervous system physiology? *J. Neurosci.* **30**, 15337–15357 (2010).
- 8. Yong, V. W. Metalloproteinases: mediators of pathology and regeneration in the CNS. *Nat. Rev. Neurosci.* **6**, 931–944 (2005).
- Greene, J. et al. Molecular cloning and characterization of human tissue inhibitor of metalloproteinase 4. *J. Biol. Chem.* 271, 30375–30380 (1996).
- Brkic, M., Balusu, S., Libert, C. & Vandenbroucke, R. E. Friends or Foes: Matrix Metalloproteinases and their multifaceted roles in neurodegenerative diseases. *Mediators Inflamm.* **2015**, 620581 (2015).
- 11. Yang, Y. & Rosenberg, G. A. Matrix metalloproteinases as therapeutic targets for stroke. *Brain Res.* **1623**, 30–38 (2015).
- Bi, J., Shan, W., Luo, A. & Zuo, Z. Critical role of matrix metallopeptidase 9 in postoperative cognitive dysfunction and agedependent cognitive decline. *Oncotarget* 8, 51817–51829 (2017).
- 13. Li, Z. et al. Surgery-induced Hippocampal Angiotensin II elevation causes blood-brain barrier disruption via MMP/TIMP in aged rats. *Front. Cell. Neurosci.* **10**, 105 (2016).
- 14. Ni, P. et al. IL-17A contributes to perioperative neurocognitive disorders through blood-brain barrier disruption in aged mice. *J. Neuroinflamm.* **15**, 332 (2018).
- Cavallari, M. et al. Longitudinal diffusion changes following postoperative delirium in older people without dementia. *Neurology* 89, 1020–1027 (2017).
- Abu-Rumeileh, S. et al. The multifaceted role of neurofilament light chain protein in non-primary neurological diseases. *Brain* 146, 421–437 (2023).
- Halaas, N. B. et al. Neurofilament light in serum and cerebrospinal fluid of hip fracture patients with Delirium. *Dement. Geriatr. Cogn. Disord.* 46, 346–357 (2018).
- Girard, T. D. et al. Associations of markers of inflammation and coagulation with delirium during critical illness. *Intensive Care Med.* 38, 1965–1973 (2012).
- 19. Sessler, C. N. et al. The Richmond Agitation-Sedation Scale: validity and reliability in adult intensive care unit patients. *Am. J. Respir. Crit. Care Med.* **166**, 1338–1344 (2002).
- Tieges, Z., McGrath, A., Hall, R. J. & Maclullich, A. M. Abnormal level of arousal as a predictor of delirium and inattention: an exploratory study. *Am. J. Geriatr. Psychiatry* **21**, 1244–1253 (2013).
- Hall, R. J., Meagher, D. J. & MacLullich, A. M. Delirium detection and monitoring outside the ICU. *Best. Pract. Res. Clin. Anaesthesiol.* 26, 367–383 (2012).
- Neerland, B. E. et al. The protocol of the Oslo Study of Clonidine in Elderly Patients with Delirium; LUCID: a randomised placebocontrolled trial. *BMC Geriatr.* **15**, 7 (2015).
- 23. Jorm, A. F. The Informant Questionnaire on cognitive decline in the elderly (IQCODE): a review. *Int. Psychogeriatr.* **16**, 275–293 (2004).
- Mayhew, D., Mendonca, V. & Murthy, B. V. S. A review of ASA physical status—historical perspectives and modern developments. *Anaesthesia* 74, 373–379 (2019).
- Idland, A. V. et al. Biomarker profiling beyond amyloid and tau: cerebrospinal fluid markers, hippocampal atrophy, and memory change in cognitively unimpaired older adults. *Neurobiol. Aging* 93, 1–15 (2020).
- Knapskog, A.-B. et al. Cerebrospinal fluid sTREM2 in Alzheimer's disease: comparisons between clinical presentation and AT classification. *Sci. Rep.* **10**, 15886 (2020).

- Medbøen, I. T. et al. Cohort profile: the Norwegian Registry of Persons Assessed for Cognitive Symptoms (NorCog)—a national research and quality registry with a biomaterial collection. *BMJ Open* 12, e058810 (2022).
- McKhann, G. M. et al. The diagnosis of dementia due to Alzheimer's disease: recommendations from the National Institute on Aging-Alzheimer's Association workgroups on diagnostic guidelines for Alzheimer's disease. *Alzheimers Dement* 7, 263–269 (2011).
- Jack, C. R. Jr. et al. NIA-AA Research Framework: Toward a biological definition of Alzheimer's disease. *Alzheimers Dement* 14, 535–562 (2018).
- Rosenberg, G. A. Matrix metalloproteinases in neuroinflammation. *Glia* 39, 279–291 (2002).
- 31. Hall, R. J. et al. CSF biomarkers in delirium: a systematic review. *Int. J. Geriatr. Psychiatry* **33**, 1479–1500 (2018).
- 32. Caley, M. P., Martins, V. L. & O'Toole, E. A. Metalloproteinases and Wound Healing. *Adv. Wound Care* **4**, 225–234 (2015).
- Davey, A., McAuley, D. F. & O'Kane, C. M. Matrix metalloproteinases in acute lung injury: mediators of injury and drivers of repair. *Eur. Respir. J.* 38, 959–970 (2011).
- Chow, A. K., Cena, J. & Schulz, R. Acute actions and novel targets of matrix metalloproteinases in the heart and vasculature. *Br. J. Pharmacol.* **152**, 189–205 (2007).
- 35. Wells, J. E. et al. An adverse role for matrix metalloproteinase 12 after spinal cord injury in mice. *J. Neurosci.* **23**, 10107–10115 (2003).
- Power, C. et al. Intracerebral hemorrhage induces macrophage activation and matrix metalloproteinases. *Ann. Neurol.* 53, 731–742 (2003).
- Martino Adami, P. V. et al. Matrix metalloproteinase 10 is linked to the risk of progression to dementia of the Alzheimer's type. *Brain* 145, 2507–2517 (2022).
- Santaella, A. et al. Inflammation biomarker discovery in Parkinson's disease and atypical parkinsonisms. *BMC Neurol.* 20, 26 (2020).
- Mlekusch, R. & Humpel, C. Matrix metalloproteinases-2 and -3 are reduced in cerebrospinal fluid with low beta-amyloid1–42 levels. *Neurosci. Lett.* 466, 135–138 (2009).
- Stomrud, E., Björkqvist, M., Janciauskiene, S., Minthon, L. & Hansson, O. Alterations of matrix metalloproteinases in the healthy elderly with increased risk of prodromal Alzheimer's disease. *Alzheimers Res. Ther.* 2, 20 (2010).
- Aksnes, M. et al. Sex-specific associations of matrix metalloproteinases in Alzheimer's disease. *Biol. Sex. Differ.* 14, 35 (2023).
- Melendez-Zajgla, J., Del Pozo, L., Ceballos, G. & Maldonado, V. Tissue inhibitor of metalloproteinases-4. The road less traveled.*Mol. Cancer* 7, 85 (2008).
- Leco, K. J. et al. Murine tissue inhibitor of metalloproteinases-4 (Timp-4): cDNA isolation and expression in adult mouse tissues. *FEBS Lett.* 401, 213–217 (1997).
- Tsai, H.-C. et al. Association of matrix metalloproteinase-9 and tissue inhibitors of metalloproteinase-4 in cerebrospinal fluid with bloodbrain barrier dysfunction in patients with eosinophilic meningitis caused by Angiostrongylus cantonensis. *Am. J. Trop. Med. Hyg.* **78**, 20–27 (2008).
- Tsai, H.-C. et al. Expression of matrix metalloproteinases and their tissue inhibitors in the serum and cerebrospinal fluid of patients with HIV-1 infection and syphilis or neurosyphilis. *Cytokine* 54, 109–116 (2011).
- Killer, M. et al. Cytokine and growth factor concentration in cerebrospinal fluid from patients with hydrocephalus following endovascular embolization of unruptured aneurysms in comparison with other types of Hydrocephalus. *Neurochem. Res.* **35**, 1652–1658 (2010).
- Fatehi Hassanabad, A. et al. Comprehensive characterization of the postoperative pericardial inflammatory response: Potential implications for clinical outcomes. *JTCVS Open* **12**, 118–136 (2022).

- Cunningham, E. L. et al. CSF Beta-amyloid 1-42 concentration predicts Delirium following elective arthroplasty surgery in an observational cohort study. *Ann. Surg.* 269, 1200–1205 (2019).
- Sulik, A., Wojtkowska, M. & Oldak, E. Preanalytical factors affecting the stability of matrix metalloproteinase-2 concentrations in cerebrospinal fluid. *Clin. Chim. Acta* **392**, 73–75 (2008).
- Watne, L. O. et al. Cerebrospinal fluid quinolinic acid is strongly associated with delirium and mortality in hip fracture patients. *J. Clin. Invest.*, **133**, e163472 (2022).
- Lorenzl, S. et al. Tissue inhibitors of matrix metalloproteinases are elevated in cerebrospinal fluid of neurodegenerative diseases. *J. Neurol. Sci.* 207, 71–76 (2003).
- Rivera, S., García-González, L., Khrestchatisky, M. & Baranger, K. Metalloproteinases and their tissue inhibitors in Alzheimer's disease and other neurodegenerative disorders. *Cell. Mol. Life Sci.* 76, 3167–3191 (2019).
- Jackson, T. A., MacLullich, A. M., Gladman, J. R., Lord, J. M. & Sheehan, B. Diagnostic test accuracy of informant-based tools to diagnose dementia in older hospital patients with delirium: a prospective cohort study. *Age Ageing* **45**, 505–511 (2016).

Acknowledgements

We thank everyone at Oslo University Hospital, Diakonhjemmet Hospital, Akershus University Hospital, and Bærum Hospital who contributed to the inclusion of patients in the hip fracture cohort. We thank the NorCog registry for access to data and biological samples for the AD group. We acknowledge all the patients who have participated in NorCog, Cognorm, or the hip fracture cohort. This work has been funded by the Norwegian Health Association (#1513, #14845, #25633, #19536), the South-Eastern Norway Regional Health Authorities (#2017095), Vestre Viken Hospital Trust and Wellcome Leap's Dynamic Resilience Program (jointly funded by Temasek Trust) (#104617). The funders had no role in the design of the study, data collection, and analysis or preparation of the manuscript. We thank Dr. Muhammad Umar Sajjad for his valuable contributions to this work.

Author contributions

M.A., F.A.C., T.B.W. and L.O.W. designed the study. M.H.S., B.E.N., I.S., R.S.E., M.M., T.H.E., K.P., A-V.I, C.T.P., R.B.O. and L.O.W. contributed to data collection. B.E.N. and L.O.W. assessed whether the hip fracture patients fulfilled the DSM-5 criteria for delirium. G.C. and E.C. provided expert advice. H.Z. performed biochemical analysis. M.A. performed the statistical analysis, created the data visualisations, and wrote the manuscript. All authors contributed to the interpretation of the results and the revision of the manuscript.

Competing interests

The authors declare the following competing interests: KP reports work with Novo Nordisk NN6535-4730 trial outside the submitted work. HZ has served at scientific advisory boards and/or as a consultant for Abbvie, Acumen, Alector, Alzinova, ALZPath, Amylyx, Annexon, Apellis, Artery Therapeutics, AZTherapies, Cognito Therapeutics, CogRx, Denali, Eisai, Merry Life, Nervgen, Novo Nordisk, Optoceutics, Passage Bio, Pinteon Therapeutics, Prothena, Red Abbey Labs, reMYND, Roche, Samumed, Siemens Healthineers, Triplet Therapeutics, and Wave, has given lectures in symposia sponsored by Alzecure, Biogen, Cellectricon, Fujirebio, Lilly, Novo Nordisk, and Roche, and is a co-founder of Brain Biomarker Solutions in Gothenburg AB (BBS), which is a part of the GU Ventures Incubator Program (outside submitted work). All other authors explicitly report no disclosures.

Ethics approval and consent to participate

The study was conducted in accordance with the Declaration of Helsinki. All data and materials were collected after informed written consent from the patient, or from the next of kin if patients were unable to consent due to cognitive impairment; next of kin have the authority to grant consent in Norway under the Act on Medical and Health Research (ACT 2008-06-20). The Regional Committee for Ethics in Medical Research in Norway has approved the multi-centre study from which hip fracture patients were included (REC Central, #19337); the Cognorm study (REC South East, #12064); and the study of inflammatory biomarkers in the NorCog material (REC South East #6897). The ethical approval encompasses the current study, including analysis of MMP and TIMP biomarkers and comparisons across cohorts.

Additional information

Supplementary information The online version contains supplementary material available at https://doi.org/10.1038/s43856-024-00558-z.

Correspondence and requests for materials should be addressed to Mari Aksnes.

Peer review information *Communications Medicine* thanks the anonymous reviewers for their contribution to the peer review of this work. A peer review file is available.

Reprints and permissions information is available at http://www.nature.com/reprints

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

© The Author(s) 2024

¹Institute of Clinical Medicine, University of Oslo, Oslo, Norway. ²Department of Geriatric Medicine, Sørlandet Hospital, Arendal, Norway. ³Department of Molecular Medicine, Institute of Basic Medical Sciences, University of Oslo, Oslo, Norway. ⁴Oslo Delirium Research Group, Department of Geriatric Medicine, Oslo University Hospital, Oslo, Norway. ⁵Department of Geriatric Medicine, Prince of Wales Hospital, Sydney, NSW, Australia. ⁶Prince of Wales Clinical School, University of New South Wales, Sydney, NSW, Australia. ⁷Department of Neuromedicine and Movement Science, Norwegian University of Science and Technology, Trondheim, Norway. ⁸Department of Geriatric Medicine, Bærum Hospital, Trondheim University Hospital, Trondheim, Norway. ⁹Department of Internal Medicine, Bærum Hospital, Vestre Viken Hospital Trust, Bærum, Norway. ¹⁰Department of Geriatric Medicine, Oslo University Hospital, Oslo, Norway. ¹¹Vestfold Hospital Trust, Norwegian National Centre for Ageing and Health, Tønsberg, Vestfold, Norway. ¹²Department of Anesthesiology, Akershus University Hospital, Lørenskog, Norway. ¹³Department of Orthopedic Surgery, Akershus University Hospital, Oslo, Norway. ¹⁶Department of Orthopedic Surgery, Akershus University Hospital, Oslo, Norway. ¹⁶Department of Neuroscience and Physiology, the Sahlgrenska Academy at University of Gothenburg, Mölndal, Sweden. ¹⁶Clinical Neurochemistry Laboratory, Sahlgrenska University

Hospital, Mölndal, Sweden. ¹⁷Department of Neurodegenerative Disease, UCL Institute of Neurology, London, UK. ¹⁸UK Dementia Research Institute at UCL, London, UK. ¹⁹Hong Kong Center for Neurodegenerative Diseases, Hong Kong, China. ²⁰Wisconsin Alzheimer's Disease Research Center, University of Wisconsin School of Medicine and Public Health, University of Wisconsin-Madison, Madison, WI, USA. ²¹Centre for Public Health, Queen's University Belfast, Belfast, UK. ²²Department of Geriatric Medicine, Akershus University Hospital, Lørenskog, Norway.